Author's response to reviews

Title: What rheumatologists in the United States think of complementary and alternative medicine: results of a national survey

Authors:

Nisha J Manek (manek.nisha@mayo.edu)
Cynthia S Crowson (crowson.cynthia@mayo.edu)
Abigale L Ottenberg (ottenberg.abigale@mayo.edu)
Farr A Curlin (fcurlin@bsd.uchicago.edu)
Ted J Kaptchuck (ted_kaptchuck@hms.harvard.edu)
Jon C Tilburt (tilburt.Jon@mayo.edu)

Version: 7 Date: 5 January 2010

Author's response to reviews:

January 5th, 2010

Melissa Norton, MD
Editor-in-Chief
BioMed Central Complementary and Alternative Medicine
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UNITED KINGDOM

RE: MS# 1513484365285956—What Rheumatologists in the United States Think of Complementary and Alternative Medicine: Results of a National Survey

We are delighted our manuscript, “What Rheumatologists in the United States Think of Complementary and Alternative Medicine: Results of a National Survey,” has been accepted for publication in BioMed Central Complementary and Alternative Medicine. We have revised the format of the manuscript to conform to the journal style as follows:

1) Author affiliations have been inserted. Each affiliation has been written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country.

Each author has been linked to their corresponding affiliations with the use of superscript numbers.

2) We have stated that all authors read and approved the final manuscript.

With very best wishes for the New Year,
Nisha Manek, MD